Evolution, Medicine, and Public Health [2021] pp. 21–35 https://doi.org/10.1093/emph/eoab041 Advance access date 27 November 2021

# Early life, life course and gender influences on levels of C-reactive protein among migrant Bangladeshis in the UK

Khurshida Begum,<sup>1</sup> Gillian D. Cooper,<sup>1</sup> Nasima Akhter,<sup>1</sup> Papreen Nahar,<sup>1,2</sup> Adetayo Kasim<sup>1,3,4</sup> and Gillian R. Bentley 1.3

<sup>1</sup>Department of Anthropology, Durham University, Dawson Building, South Road, Durham DH1 3LE, UK; <sup>2</sup>Department of Global Health and Infection, University of Sussex, BSMS Teaching Building, Brighton BN1 9PX, East Sussex, UK; <sup>3</sup>Durham Research Methods Centre, Faculty of Social Sciences & Health, Durham University, Arthur Holmes Building, Durham DH1 3LE, UK and <sup>4</sup>UCB Pharmaceuticals, 216 Bath Road, Slough SL1 3WE, UK \*Corresponding author. Department of Anthropology, Durham University, Dawson Building, South Road, Durham DH1 3LE, UK. Tel: 011 44 191 334 1114; E-mail: g.r.bentley@durham.ac.uk Received 21 April 2021; revised version accepted 18 November 2021

#### ABSTRACT

**Background and objectives:** Humans co-evolved with pathogens, especially helminths, that educate the immune system during development and lower inflammatory responses. The absence of such stimuli in industrialized countries is associated with higher baseline levels of C-reactive protein (CRP) among adults who appear at greater risk for inflammatory disorders. This cross-sectional study examined effects of early life development on salivary CRP levels in 452 British-Bangladeshis who spent varying periods growing up in Bangladesh or UK. We also analyzed how gender and central obesity modulate effects on CRP. We hypothesized that: (i) first-generation Bangladeshis with higher childhood exposure to pathogens would have chronically lower CRP levels than second-generation British-Bangladeshis; (ii) effects would be greater with early childhoods in Bangladesh; (iii) effects by gender would differ; and (iv) increasing obesity would mitigate early life effects.

**Methodology:** Saliva samples were assayed for CRP using ELISAs, and anthropometric data collected. Participants completed questionnaires about demographic, socioeconomic, lifestyle and health histories. Data were analyzed using multiple linear regression.

**Results:** First-generation migrants who spent early childhoods in mostly rural, unhygienic areas, and moved to UK after age 8, had lower salivary CRP compared to the second-generation. Effects differed by gender, while waist circumference predicted higher CRP levels. CRP increased with years in UK, alongside growing obesity.

© The Author(s) 2021. Published by Oxford University Press on behalf of the Foundation for Evolution, Medicine, and Public Health. This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.



RESEARCH ARTICLE

## EVOLUTION, Medicine, & Public Health

**Conclusions and implications:** Our study supports the hypothesis that pathogen exposure in early life lowers inflammatory responses in adults. However, protective effects differed by gender and can be eroded by growing obesity across the life course which elevates risks for other inflammatory disorders.

Lay Summary: Migrants to the UK who spent early childhoods in less hygienic environments in Bangladesh that help to educate their immune systems had lower levels of the inflammatory marker, C-reactive protein (CRP) compared to migrants who grew up in UK. Both gender and increasing obesity were associated with increased levels of CRP.

KEYWORDS: C-reactive protein; inflammation; early life development; gender; Bangladeshi migrants; obesity

#### BACKGROUND AND OBJECTIVES

C-reactive protein (CRP) is released from hepatocytes as part of the inflammatory and immune response and is regularly used as a non-specific marker of systemic inflammation [1–3]. Elevation of CRP levels can take two forms: a rapid, acute rise usually associated with a current infection or injury, or a persistent, low-grade elevation linked to conditions including obesity [4], especially visceral adiposity [5, 6], metabolic disorders, including cardiovascular diseases (CVDs) [4, 7] and type 2 diabetes mellitus (T2DM) [8, 9]. CRP is thus a useful biomarker for acute and chronic inflammatory responses [10–12], using either high sensitivity assays for plasma levels, where venepuncture is possible [1, 13] or less sensitive assays for measurement of salivary levels. The latter are easier and less invasive to collect in non-clinical settings [14], and correlate well with plasma levels [15].

Accumulating evidence of low baseline levels of CRP from small-scale populations practicing traditional lifeways suggests that the chronically elevated levels typically seen in contemporary, industrialized societies may constitute an evolutionary mismatch. The underlying theory for this mismatch relates to the 'Old Friends' or 'Hygiene Hypothesis' which argues that a reduction of immunological stimuli, particularly from soil-transmitted helminths (STH), has increased vulnerability to autoimmune diseases such as asthma, allergies and Crohn's Disease [16-19]. Specifically, it is argued that the human immune system is highly plastic and early life experiences form and educate the adult immunological phenotype [20-23], including establishing effective T-cell and anti-inflammatory regulatory networks [17, 22, 23]. Increasing evidence also supports epigenetic mechanisms underpinning different inflammatory phenotypes [24, 25].

Bangladesh, as a semi-tropical, low- to middle-income country (LMIC) in South Asia, still has a high prevalence of infectious and parasitic diseases [26]. Many people lack access to clean water and sanitary facilities, while the quality of the health care system requires improvement [27, 28]. Helminth infections have remained high across the country in both rural and urban locations and across socioeconomic groups despite national, anti-helminthic policies since 2008 [29–31]. Drawing upon accumulating evidence for lower baseline levels of CRP among populations in similar environments [1, 32, 33], one might therefore predict that Bangladeshi adults, particularly those who grew up in rural areas and were exposed to STH during childhood, would have chronically lower levels of CRP.

Bangladesh is primarily a Muslim country with strongly gendered role expectations, particularly among the more affluent classes, which might also affect exposure to STH. In addition, many Bangladeshis, regardless of wealth, are now experiencing the dual burden of morbidity where a growing proportion are developing non-communicable diseases (NCDs) including obesity, T2DM and CVDs characterized by chronic inflammation [34, 35]. Such characteristics also hold for Bangladeshi migrants to Europe and the North American continent [36–38].

Bangladeshis in the UK, who primarily originate from the more affluent middle classes who own land in their home country and have household helpers, have had a long history of migration with third-generation children now growing up. This offers the opportunity to investigate aspects of health across generations who spent their childhoods in markedly different conditions. First-generation migrants would have experienced various portions of their lives in contrasting environments permitting examination of the effects of the childhood environment and differential exposures to pathogens on the adult phenotype, as well as effects of changing environments on characteristics associated with inflammatory disorders.

In this paper, we therefore tested the hypothesis that the environment in early life would influence levels of salivary CRP in adulthood using data from a cross-sectional study of migrant Bangladeshis in the UK. We predicted that those who were born and grew up in Bangladesh, with exposure to greater pathogen loads including STH, would have chronically lower levels of salivary CRP as adults compared to Bangladeshi migrants who spent their developmental years growing up in the UK. Second, following our earlier findings of differential effects on reproductive hormone levels in migrant adults, depending on whether they moved to the UK before or after the age of 8 [39-42], we tested the hypothesis that adult levels of salivary CRP would differ among child migrants depending on their age at migration. Specifically, we predicted that levels of CRP would be lower among child migrants who moved to the UK after 8 years of age, matching those of adult migrants, while we also predicted that levels of CRP among child migrants who moved before or at age 8 would more closely match those of the secondgeneration. Third, given recent discussions of how studies of CRP have paid insufficient attention to gender differences in reporting CRP levels, we tested the hypothesis that CRP levels and the effects of early life environments would differ by gender. We predicted that first-generation women would have higher levels of CRP owing to lower exposures to STH. Finally, we tested the hypothesis that lifestyle changes in the UK predisposing individuals to obesity would influence levels of CRP, predicting that increased markers of overweight/obesity would elevate levels of CRP in adulthood, and again that women would be more vulnerable to these effects.

#### **METHODOLOGY**

#### **Participants**

The study used cross-sectional data from a collaborative project designed to investigate relationships between migration and health among 562 first- and second-generation, Bangladeshi migrants living either in northeast England (n = 275) or in London (n = 287), and aged 25–40 years when age-related changes to CRP would be expected to be minimal [43, 44]. A subset of individuals (n = 479, 85%) provided saliva samples for hormonal analyses, from which a further seven participants were excluded due to inadequate or poor-quality samples, leaving a pool of 468 participants. In addition, 16 individuals with levels of salivary CRP >3000 pg/ml, above the limits of assay detection,<sup>1</sup> and likely indicative of an acute infection, were removed from the dataset, leaving a final sample of 452 for this study.

Data were collected between May 2013 and April 2015. Subjects were categorized into four groups: (i) first-generation, adult migrants, born and brought up in Bangladesh, who migrated to the UK as adults aged >18 (n=174); (ii) first-generation, child migrants born in Bangladesh who came to the UK aged between 9 and 18 (n=73); referred to henceforth as Child Migrants >8); (iii) first-generation, child migrants born in Bangladesh who migrated to the UK aged  $\leq 8$  years (n=74); referred to henceforth as Child Migrants  $\leq 8$ ); and (4) second-generation British-Bangladeshis who were born and brought up exclusively in the UK, with parents of Bangladeshi origin (n=129). For many analyses, Groups 2 and 3 (all Child Migrants who migrated to the UK  $\leq 18$  years, n=149) were combined into one, making only three groups altogether.

Exclusion criteria for the study were thyroid conditions, type I diabetes, psychosis, bipolar disorder or major depression. We also excluded women who were currently either pregnant or lactating due to the potential effect of these conditions on salivary

hormone levels. In both London and northeast England, we used local Bangladeshi residents as recruiters while, in the northeast, we also employed local Bangladeshi residents as research assistants who helped with data collection including interviews. The latter were trained prior to beginning work and monitored throughout the project to ensure consistency and quality of data collection.

All participants answered a structured questionnaire that requested information about demographic and migration histories, socioeconomic, educational and environmental backgrounds, stress, health histories both recent and past, use of any current medications, dietary intake and physical exercise. Health histories included a question about any illness episodes in the last two weeks. Questionnaires were translated into Bengali and independently back-translated into English to check for accuracy. Participants were given the option of answering questionnaires in Bengali or English, but only 12 subjects in London and two in the northeast chose the Bengali questionnaire. The questionnaires were piloted extensively in draft form and modified following feedback from participants.

Anthropometric data were collected using standardized techniques [45]. Height was measured to the tenth of a cm (0.1 cm) using a portable stadiometer, weight to the tenth of a kg (0.1 kg) (according to manufacturer's guidelines) using a portable weighing scale, and waist circumference (WC) to the tenth of a cm (0.1 cm) using a measuring tape. The body mass index (BMI) was calculated from these data. Levels of overweight and obesity conform to World Health Organisation (WHO) standards [46], while central obesity was determined using the UK National Institute for Clinical Excellence (NICE) criteria for South Asians (>0.94 cm for males and >0.80 for women) [47].

Participants followed detailed written and verbal instructions to collect and return saliva samples (collected thrice daily) and were asked to store samples in the fridge once collected. Subjects were requested to refrain from tooth brushing, betel nut chewing, eating and drinking for one hour before saliva sampling since betel nut use can be common among older first-generation adults and has been shown to interfere with other saliva sampling regimes [48]. Samples were posted to the laboratory in pre-paid envelopes the day after collection using next day delivery. On receipt at the laboratory, they were immediately frozen at  $-20^{\circ}$ C until the day of assay. Collection and arrival dates were both logged.

#### **CRP** analysis

Evening saliva samples were used to analyze CRP as the most recent relative to time spent in transit. All samples were assayed at Durham University's Endocrinology and Ecology laboratory using commercially available salivary CRP enzyme immunoassay (ELISA) kits supplied by Salimetrics<sup>®</sup> and following

<sup>1</sup> Salimetrics, the company that supplied the CRP ELISA kit have since introduced a new assay but instructions and specifications for the assay used at the time of analyses can be found at: https://fnkprddata.blob.core.windows.net/domestic/data/datasheet/SAL/1-3302. pdf

manufacturer's recommended guidelines (see footnote 1). Samples were randomized across well plates to ensure all participant groups (generation, gender and region) were represented equally. If a sample was inadequate for assay or produced a value outside the calibrator range it was excluded.

On the day of assay, samples were defrosted and centrifuged (3500 rpm  $\times$  20 min at 4°C) to facilitate the separation of mucins and oral debris from the clear saliva fraction, which was aliquoted ready for assay. Saliva samples were diluted (×10) prior to assay as instructed by the manufacturer. Each saliva sample was assayed in duplicate at a volume of 50 µl per well and according to the supplied protocol. The sensitivity of the CRP assay was 10 pg/ml with a calibrator range of 93.75–3000 pg/ml. The intra- and inter-assay CVs were 8.8% and 9.1%, respectively.

#### Statistical analysis

CRP data were log-transformed prior to analyses. Tertiles of CRP were also created using data for geometric mean levels. Both the categorical variable, 'Generation' and the numerical variable, 'Number of Years in Bangladesh', were used as proxy measures of exposure to environmental differences in early life. Variables were analyzed descriptively, while bivariate Pearson correlations examined initial associations between variables, Chi-square tests were used to look for differences between generations for categorical variables and *t*-tests or ANOVA for continuous variables.

Univariate analyses explored relationships between logCRP and potential demographic covariates including Generation, Age, Marital Status (married or partnered, single or separated, divorced, widowed), Gender, Number of Years Lived in Bangladesh or the UK and whether individuals grew up in a rural or urban environment. We also explored relationships with anthropometric measures (BMI, WC), measures of health (scores from the General Health Questionnaire-12) [49], physical activity (walking at least 20 min daily), socioeconomic status and an acculturation score derived from a modified Suinn-Lew Asian Acculturation Scale [50]. Variables were included in multivariate regression models if they reached a significance of 0.1. Some variables were highly significantly correlated (Age and Generation; BMI and WC; Generation and Growing Up in a Rural or Urban Location; and Number of Years Lived in Bangladesh with Number of Years Lived in the UK, both of which were also highly correlated with Age, P < 0.001).

The final list of covariates included in four different multiple linear regression (MLR) models were therefore Generation, Number of Years Lived in Bangladesh, and Number of Years Lived in the UK; Gender and WC appeared as additional covariates in each of the models. An initial MLR (Model 1) explored the relationship between logCRP and the categorical variable Generation (here, as three groups: Adult Migrants, All Child Migrants and the Second-Generation), and included Gender and WC as additional covariates. A second (Model 2) duplicated the first, but used the covariate Generation comprised of four groups, where Child Migrants were split into those who migrated >8 or <8 years. We also ran additional models using centered interaction terms to check for the potential role of mediators and moderators. A third model used the predictors, Number of Years Spent in Bangladesh, Gender and WC (Model 3a), and included an interaction term between Number of Years Spent in Bangladesh and WC, again with centered predictor variables (Model 3b). A final model used Number of Years Lived in the UK to be able to compare the potential effect of this variable separately, plus Gender and WC (Model 4). All statistics were performed using SPSS version 27, and P-values <0.05 were considered significant.

#### Ethics

Ethical permission for the Project was granted by the Ethics Committee of the Department of Anthropology, Durham University. All participants gave informed consent, were provided with information about the Project in either English or Bengali, as appropriate, and were compensated for their time and effort. All subjects were anonymized for data entry, and data were stored in accordance with GDPR requirements.

#### RESULTS

Mean age of participants was  $32.9 \pm 4.8$  years and differed significantly by generation (ANOVA;  $F_{2,449}$ =28.863, P<0.001); second-generation participants were significantly younger due to the history of Bangladeshi migration (Table 1). The majority of adult migrants (70%) had grown up in a rural location compared to 23% for child migrants overall, and only 2% for the second-generation (Table 1). Slightly more than half the subjects (54%) were either overweight (39%) or obese (15%). Individuals characterized by general obesity differed by generation with greater proportions among the Child Migrants < 8(22%) and the Second-Generation (19%) compared to Adult Migrants (13%) and Child Migrants >8 (8%; Table 1), but these differences were not significant (ANOVA,  $F_{3,448}$  =0.339, P > 0.05). While both men (41%) and women (40%) were equally overweight, significantly more women (20%) were obese compared to their male counterparts (10%; t = -2.509, P = 0.012).

A high proportion (55%) of the population also had a WC above the National Institute for Clinical Excellence (NICE, UK) cut-off for central obesity. Prevalence of central obesity differed across generations (Adult Migrants: 53%; Child Migrants >8: 45%; Child Migrants  $\leq$ 8: 58%; Second-Generation: 61%;



#### Table 1. Descriptive statistics by generation

Variable <sup>a</sup>		Total ( <i>n</i> = 452)			
	ADU (n = 175)	CHI > 8 (n = 73)	CHI < 8 (n = 74)	2ND (n=129)	
$Age^{\dagger}$ (years)	$33.7 \pm 4.6$	$34.3 \pm 4.6$	34.1 ± 4.6	$\textbf{30.4} \pm \textbf{4.4}$	$\textbf{32.9} \pm \textbf{4.8}$
Gender <sup>*</sup>					
Male	88 (50.3)	37 (50.7)	23 (31.1)	52 (40.0)	200 (44.2)
Female	87 (49.7)	36 (49.3)	51 (68.9)	78 (60.0)	252 (55.8)
BMI	$25.7 \pm 3.5$	$\textbf{25.8} \pm \textbf{4.8}$	$\textbf{26.3} \pm \textbf{4.9}$	$26.4 \pm 5.0$	$\textbf{26.0} \pm \textbf{4.4}$
Obesity <sup>c</sup>					
Underweight	4 (2.3)	2 (2.7)	3 (4.1)	3 (2.3)	12 (2.7)
Normal	72 (41.4)	31 (42.5)	33 (45.2)	56 (43.4)	192 (42.8)
Overweight	76 (43.7)	34 (46.6)	21 (28.8)	46(35.7)	177 (39.4)
Obese	22 (12.6)	6 (8.2)	16 (21.9)	24 (18.6)	68 (15.1)
Waist circumference	$89.3\pm8.5$	$87.0 \pm 9.9$	$\textbf{88.6} \pm \textbf{11.4}$	$89.5 \pm 11.8$	$88.9 \pm 10.3$
Central obesity <sup>d</sup>					
Normal	82 (46.9)	40 (54.8)	31 (41.9)	51 (39.2)	204 (45.1)
Obese	93 (53.1)	33 (45.2)	43 (58.1)	79 (60.8)	248 (54.9)
Grew up <sup>†</sup>					
Rural	123 (70.3)	25 (35.7)	6 (8.6)	2 (1.5)	156 (35.1)
Urban	52 (29.7)	45 (64.3)	64 (91.4)	128 (98.5)	289 (64.9)
CRP $^{e}$ (pg/ml),** mean $\pm$ SD	$293 \pm 303$	$315\pm475$	$300\pm314$	$441\pm574$	$340\pm430$

 $^{*}P < 0.05; ^{**}P < 0.01; ^{\dagger}P < 0.001.$ 

<sup>a</sup>Presented as mean  $\pm$  SD for numerical data and numbers (%) for categorical data.

<sup>b</sup>ADU = adult migrant; CHI > 8 = child migrants who moved aged over 8;  $CHI \le 8 = child$  migrants who moved under or equal to age 8; 2ND = Second-Generation.

<sup>c</sup>Defined according to WHO [46] criteria as 'underweight' <18.5; 'normal' = 18.5–24.9; 'overweight' = 25.0–29.9; 'obese' = >30.0. <sup>d</sup>Defined as >80 cm for women and >94 for men [47].

<sup>e</sup>Median CRP for ADU =182.6 pg/ml, CHI >8 = 163.0 pg/ml, CHI <8 = 205.8 pg/ml, 2ND= 225.5 pg/ml, Total = 192.5 pg/ml.

Table 1), but these differences were not significant (ANOVA,  $F_{3,451} = 1.060, P > 0.05$ ). Significantly more women (75%) than men (29%) were characterized by central obesity (t = 2.903, P = 0.004). Both BMI (r = 0.172, P = 0.02) and WC (r = 0.287, P < 0.001) were significantly positively correlated with age among the Adult Migrants, but were not significantly associated with either of the child migrant groups or the Second-Generation (P > 0.05). Both BMI and WC increased with time spent in the UK following migration among the First Generation, as did levels of logCRP (Supplementary Figs S1 and S2), but BMI was significantly correlated with Years Lived in the UK only among the Adult Migrants (r = 175, P = 0.021). Profiles for Child Migrants >8 looked more similar to the Adult Migrants, while Child Migrants  $\leq$ 8 years matched more closely to the Second-Generation (Supplementary Figs S1 and S2).

Mean salivary CRP levels for all participants was 340 (SD  $\pm$  430) pg/ml, but levels differed significantly by generation ( $F_{3,451}$ =3.380, P=0.018, Table 1, Fig. 1). When comparing CRP tertiles by central obesity (yes/no) across generations, there

were significant differences among those with a normal WC ( $\chi^2 = 6$ , n = 204, P > 0.007), but none between generations for those who fell into the category of being centrally obese (n = 248; Supplementary Table S1). When comparing tertiles of CRP by BMI categories across generations, those with a normal BMI were close to significance ( $\chi^2 = 6$ , n = 204, P = 0.06), but the results were not significant among those who were overweight/obese (n = 245; Supplementary Table S1).

There was no significant correlation between Age and logCRP levels for either the Adult Migrants (r=0.131, P=0.09), Child Migrants >8 (r=-0.86, P=0.47) or Child Migrants  $\leq 8$  (r=0.47, P=0.69), but the correlation was significant and negative for the Second-Generation (r=-0.209, P=0.02). LogCRP was significantly positively correlated with Number of Years Lived in the UK (r=0.095, P<0.05), and significantly negatively correlated with Number of Years Lived in the UK (r=0.098). There were no significant associations between logCRP and individuals walking more than 20 min each day, self-assessed financial situation or net household income.



### Box Plot of CRP (log<sup>10</sup>) by Generation

Figure 1. Box plot of CRP (log10) by generation

Unadjusted Model 1, evaluating whether Generation predicted logCRP levels, and where Child Migrants formed one group, was highly significant ( $r^2 = 0.021$ ;  $F_{2,451} = 4.801$ , P = 0.009). It showed that Adult (t = -2.728, P = 0.007) and Child (t = -2.736, P = 0.006) Migrants had significantly lower logCRP levels compared to the Second-Generation (Table 2). After adjusting for Gender and WC, the Model was still highly significant ( $r^2 = 0.092$ ;  $F_{4,451} = 11.323$ , P = <0.001) and significant differences remained between Adult Migrants and the Second-Generation (t = -2.400, p = 0.017) as well as Child Migrants and the Second-Generation (t = -2.532, P = 0.012; Table 2). The adjusted model also revealed that females (t = 4.524, P <0.001) had significantly higher logCRP compared to males, while logCRP increased significantly with a higher WC (t = 4.373, P < 0.001; Table 2).

For the unadjusted results in Model 2, Generation (using four groups) again significantly predicted variation in logCRP levels ( $r^2 = 0.02$ ;  $F_{3,451} = 3.041$ , P = 0.03). Both Adult Migrants (t = -2.562, P = 0.01) and Child Migrants >8 (t = -2.585, P = 0.01) were associated with significantly lower logCRP levels compared to the Second-Generation, while Child Migrants  $\leq 8$  were not (t = -1.470, P = 0.142; Table 2). After adjusting for Gender and WC, the model remained highly significant ( $r^2 = 0.089$ ,  $F_{3,451} = 8.742$ , P < 0.000). Again, both Adult Migrants  $\geq 8$  (t = -2.005, P = 0.05) were significant predictors in the model relative to the Second-Generation, but not Child Migrants  $\leq 8$  (t = -1.649, P > 0.05), while both Gender (t = -4.424, P < 0.005).

0.001) and WC (t = 4.365, P < .001) were highly significant predictors of logCRP levels (Table 2). Figure 2 reveals that individuals with central obesity looked similar in their logCRP levels regardless of Generation, while it was those with a normal WC who reflected generational, early life differences in logCRP levels.

We further explored the potential role of both WC and Gender as either mediators or moderators of the effect of early years spent in Bangladesh proxied by Generation. In relation to the former, WC was not significantly correlated with Generation (r=0.01, P = 0.831), suggesting it is not a mediator. Furthermore, a model where a centered interaction term between WC and Generation was added to the MLR using just these two variables as predictors for logCRP levels was significant overall ( $r^2 = 0.064$ ,  $F_{7,451} = 4.344$ , P < 0.000). However, the interaction removed any significance for WC in the main model (t=1.098, P=0.273), while retaining significance for Adult Migrants (t = -2.711, P = 0.007) and Child >8 (t = -2.505, P = 0.013). The only significant interaction term was WC \* Adult Migrants (t = 2.414, P = 0.016), suggesting WC is a weak moderator at best, despite this model being significant overall (Supplementary Table S2, Model S1).

In relation to Gender, this variable was significantly correlated with Generation (r = 0.111, P = 0.018). Gender also acted as a highly significant predictor of logCRP in a model with just Generation (four groups) as an additional covariate ( $r^2 = 0.050$ ,  $F_{4,447} = 5.925$ , P < 0.000) (Supplementary Table S2, Model S2). These results suggest Gender acts as a partial mediator of the

	Unadjusted Estimate			Adjusted Estimate		
Model 1	В	95	5% CI	В	95% CI	
Variables	Unadjusted <sup>**</sup>			Adjusted <sup>†</sup>		
Generation						
Adult migrants	-0.131**	-0.22	26, -0.037	-0.112*	-0.204, -0.020	
Child migrants	-0.136**	-0.23	34, -0.038	-0.122*	-0.217, 0.027	
Second-generation		Ref		Ref		
Female				0.174 <sup>†</sup>	0.098, 0.249	
Male		Ref				
Waist circumference				0.008 <sup>†</sup>	0.004, 0.012	
Model 2						
Variables	Unadjusted <sup>*</sup>			$Adjusted^\dagger$		
Generation						
Adult migrants	-0.123 <sup>**</sup>	-0.2	.17, -0.029	-0.103*	-0.195, -0.012	
Child migrants $> 8$	-0.157**	-0.2	.76, -0.038	-0.118*	-0.234, -0.002	
Child migrants < 8	-0.089	-0.	207, 0.030	-0.096	-0.211, 0.018	
Second-generation		Ret		R	et	
Female				0.172	0.095, 0.248	
Male Waist circumference		Ref		0.008 <sup>†</sup>	0.004, 0.012	
Model 3a Variables		Unac	liusted**	۵d	iusted <sup>†</sup>	
Vallables				, (0)		
Number of years lived in Ba	ingladesh	-0.005	-0.009, -0.001	-0.004^	-0.008, 0.000	
Female				0.165'	0.088, 0.241	
Male			Ref	0.000 <sup>+</sup>	0.005 0.010	
Waist circumference				0.008	0.005, 0.012	
Model 3b						
Variables			$Unadjusted^\dagger$	Ac	$ljusted^\dagger$	
Number of years lived in Ba	Ingladesh	-0.005*	-0.009, -0.00	-0.005 <sup>*</sup>	-0.008,-0.001	
, Female Male	0		Ref	0.162 <sup>†</sup>	0.086, 0.237	
Waist circumference				0.010 <sup>†</sup>	0.006, 0.014	
WC by number of years li	ved in Banglades	h		0.001**	0.000, 0.001	
Madel 4						
Variables	Unadjusted*		$Adjusted^\dagger$			
Number of Years Lived in th	ne UK	0.004*	0.000. 0.008	0.002	-0.001. 0.006	
Female				0.171 <sup>†</sup>	0.094. 0.248	
Male		R	ef		0.001, 0.270	
Waist circumference				0.008	0.005 0.012	

 $^{*}P < 0.05; \ ^{**}P < 0.01; \ ^{\dagger}P < 0.00.$ 



Figure 2. Boxplots of CRP (log10) by generation, gender and waist circumference

effects of Generation on logCRP. When an interaction term between Gender and Generation was added to this regression model, the model remained highly significant ( $r^2 = 0.064$ ,  $F_{7,444} = 4.368$ , P < 0.000), but Gender was no longer significant in the main part of the model (t = -0.114, P = 0.909). The interaction terms for Gender \* Adult Migrants (t = 2.170, P = 0.031) and Gender \* Child >8 (t = 2.122, P = 0.034) were significant, while Gender \* Child  $\leq 8$  was not (t = 1.709, P = 0.088; Supplementary Table S2, Model S2).

Given these results, we then ran Models 1 and 2 again separately by Gender including WC as a covariate along with Generation (Supplementary Table S3, Models S3a, S3b, S4a and S4b). Adjusted Model S3a, selecting for males only (n=200) and using three groups for Generation, was significant overall ( $r^2$ =0.104,  $F_{3,196}$ =7.609, P < 0.001) with Adult Migrants (t=-2.879, P = 0.004), Child Migrants (t=-3.049, P=0.003) and WC (t=2.975, P=0.003) all as significant covariates. Results for just males were similar for Model S4a (using four groups for Generation) with the adjusted Model being significant overall ( $r^2$ =0.105,  $F_{4,195}$ =5.691, P < 0.001), and Adult Migrants (t=-2.873, P=0.005), Child Migrants > 8 (t=-2.775, P=0.006), Child Migrants ≤ 8 (t=-2.177, P= 0.031) and WC (t=2.943, P=0.004) as significant covariates. However, when we selected for just women (n=252) in the dataset, both Models S3b ( $r^2 = 0.039$ ,  $F_{3,248} = 3.328$ , P < 0.02) and Model S4b ( $r^2 = 0.037$ ,  $F_{4,247} = 2.381$ , P < 0.052) were significant overall, WC remained a significant covariate (Model 4a: t = -3.051, P = 0.003; Model 4b: t = -3.057, P = 0.002), but neither Adult Migrants or Child Migrants were significant predictors.

Unadjusted results for Model 3a (Table 2), using Years Lived in Bangladesh as the dependent variable, showed that this Model was highly significant ( $r^2 = 0.016$ ,  $F_{1,450} = 7.096$ , P <0.001) and revealed a significant relationship between this variable and logCRP (t = -2.664; P = 0.008). The adjusted Model was still highly significant ( $r^2 = 0.086$ ,  $F_{3.448} = 14.053$ , P < 14.0530.001), and Years Lived in Bangladesh remained significant (t = -2.197; P = 0.029), while Gender (t = 4.292; P < 0.000), and WC (t = 4.572; P < 0.000) were likewise significant covariates. In Model 3b, which added a centered interaction term between Years Lived in Bangladesh and WC, the model was still highly significant ( $r^2 = 0.103$ ,  $F_{4.447} = 12.892$ , P < 0.001) with significant covariates of Years Lived in Bangladesh (t = -2.521, P = 0.01), Gender (t = 4.193, P < 0.000) and WC (t = 5.202, P < 0.000). The interaction term was also significant (t = 2.956, P = 0.003).

Finally, Model 4 used Years Lived in the UK as a predictor variable (Table 2). The unadjusted model was significant

 $(r^2 = 0.009, F_{1,450} = 4.137, P = 0.043)$  and showed a significant relationship between logCRP and the independent variable (t = 2.034, P = 0.043), with more Years in the UK associated with higher levels of logCRP. When the Model was adjusted for Gender and WC, the Model was highly significant  $(r^2 = 0.079, F_{3,448} = 12.858, P < 0.001)$ , while Years Lived in the UK was no longer significant (t = 1.251, P > 0.05), but both Gender (t = 4.384, P < 0.000) and WC (t = 4.439, P < 0.000) were highly significantly associated with logCRP.

#### CONCLUSIONS AND IMPLICATIONS

In recent years, the association between degrees of exposure to infections in early life and persistent low-grade inflammation has been increasingly studied, particularly in relation to risks for chronic diseases in later life [1, 2, 32]. The argument has been that education of the immune system during early life and frequent exposure to pathogens serves to regulate activation and deactivation of the immune system and inflammatory responses [1, 20–23, 51, 52]. Individuals who lack such exposures may be unable to downregulate inflammation effectively, potentially explaining chronically elevated levels of CRP among many individuals who develop in environments with lower microbial exposures sures typical of highly industrialized societies [2, 53].

McDade et al. [1] for example, measured levels of CRP using a high specificity plasma assay among participants in the Cebu Longitudinal Health and Nutrition Survey in the Philippines where infectious and parasitic diseases are still high compared to northern hemisphere countries. They found that elevated levels of microbial exposures in childhood were associated with chronically lower levels of CRP as an adult. Similarly, low chronic levels of CRP were found among the indigenous Shuar of Ecuador who practise a mixed subsistence strategy of horticulture, fishing and hunting, and who are exposed to high levels of diseases in their tropical, Amazonian environment [2, 33]. Both of these studies focused on populations that remained in their natal environments as adults.

A more recent study, using dried blood spots, examined CRP levels from a small sample (n = 31) of Ecuadorian migrants in the USA who originated from the same region as the Shuar and who had migrated mostly as adults (n = 28), defined in this study as aged >13 [32]. While examining the hypothesis that higher exposure to pathogens in early life would lower baseline levels of CRP among these migrants, the authors also took serial measures of CRP to measure inter-individual variability across a period of two months. As predicted, median levels of CRP in these migrants were low relative to other populations, while both BMI and gender predicted CRP levels in multiple variable regression models.

Here, in our paper, we used a larger, multi-generational sample of 452 British-Bangladeshis, first testing the hypothesis and Begum et al. | 29

associated prediction that early life exposure to greater immunological challenges would lower adult levels of salivary CRP and that there might be a particular period in early life which had greater effect. We also explored the role of gender in influencing early life exposures and logCRP predicting that women would have higher levels. Since migrant Bangladeshis in the UK and other industrialized countries are at increasing risk of inflammatory disorders such as overweight/obesity, T2D and CVDs [54-56], we predicted that more years spent in the UK would increase levels of CRP. Our hypotheses and predictions were supported: those individuals who had grown up in Bangladesh had lower levels of logCRP with greater effects observed if they had spent their early childhood (before age 8) in that country. Furthermore, gender appeared to mediate these effects in women who, as a group, did not show a significant impact of the early life environment on levels of logCRP and who were also more vulnerable to obesity across the life course. In general, spending more years in the UK with accumulating levels of obesity increased levels of inflammation.

Although we do not have direct measures of STH exposure among the study participants, the majority of first-generation migrants (77%) had grown up in rural areas of Bangladesh where evidence suggests they would have been exposed to a wide range of infectious and parasitic diseases compared to second-generation British-Bangladeshis who grew up in the UK [28, 57]. Participants in the study were aged 25-40 at recruitment, meaning most would have been children during the late 1970s and into the 2000s, depending on their age at recruitment. During those decades, 96% of the rural Bangladeshi population had access to tubewells for drinking purposes, but hygiene and sanitation in the country were far less developed relative to today [58]. For example, in 1985, only 4% of the rural population had access to a sanitary latrine, rising to 26% in 1991, while 35% had unhygienic latrines (e.g. uncovered pits, 'bucket' or open latrines over water), and 39% of people used open defecation [58]. The prevalence of STH in Bangladeshi children also remained high up until 2005 when a survey estimated that 80% of those examined were infected with at least one kind of STH [31]. In a more recent study [59], half the people tested in Bangladesh were still positive for STH, while the rate was up to 90% among those living in the worst conditions.

Furthermore, children were still not widely vaccinated against measles, tetanus, polio, diphtheria, whooping cough and tuberculosis until the Bangladeshi government launched the WHO Expanded Programme of Immunization in 1979, raising country-wide levels of vaccination from 2% in 1984, to 81% in 2000 and >95% in 2019 [60, 61]. Associated with these statistics were estimates that 35% of child mortality in Bangladesh was attributable to diarrheal diseases in 1991, reflecting unhygienic environments and exposure to pathogenic microbes [58]. Moreover, a recent study of drinking water quality in Bangladesh, using data from 2012 to 2013, still showed medium levels of contamination in households owning chickens and/or cattle that would likely have exposed individuals to diarrheal diseases [62]. Matters would have been worse in the last millennium.

In our regression models, both Generation (Models 1 and 2) and Number of Years Lived in Bangladesh (Model 3) were significant predictors of logCRP levels, with lower CRP among individuals who had spent their childhoods in Bangladesh, when presumably exposures to pathogens might have had a critical impact on the developing immune system, particularly in early childhood. Specifically, child migrants who had spent their first eight years in Bangladesh aligned with the adult migrants when predicting lower logCRP levels, while child migrants who moved to the UK before age 8, were more similar to the secondgeneration British-Bangladeshis. In contrast, the number of years spent in the UK did not significantly predict logCRP levels when accounting for gender and WC.

These results lend support to accumulating evidence from studies in evolutionary medicine that humans co-evolved with STH and that such exposures, especially during early life, modulate inflammatory responses [1, 2, 32, 33]. Two unanswered questions, however, have been: (i) how gender might be a significant factor in differentiating among these early life and life course exposures; and (ii) the extent to which life course exposures to other inflammatory conditions, including overweight and obesity, could override the potentially protective effects of a better-educated immune response acquired during development.

In relation to the first point concerning gender, Klein and Flanagan [63] have recently discussed the absence of reporting results by sex in studies of immunological function (amounting to <10% in the literature), sex referring to biologically based differences between females and males. Similarly, Clough [64] noted that proponents of the Hygiene Hypothesis have not paid sufficient attention to gender variation that might result in different vulnerabilities to STH and resulting health across the life course. This also appears to be replicated in some of the more anthropological studies relating to CRP levels among different populations [1, 32, 33]. Here, gender refers to differences in both culture and behavior that reflect specific expectations placed on girls/boys and women/men.

Both Gender and WC were highly significant covariates in Models 1–3 and increased the ability to predict logCRP by 7% in adjusted Models 1 and 2. Mediation analyses of both variables and their effects on Generation as predictors of logCRP levels in Model 2 showed that Gender is a partial mediator of logCRP levels and that WC may be a weak moderator. Higher CRP levels are known to be related to greater central adiposity [65], and this is a particular risk factor for women [5, 66], as well as South Asians [4]. The women who participated in the research here were much more likely to be obese than men, and a higher proportion were also characterized by central obesity [38]. It may be that Bangladeshi women are much more susceptible to an increase in overweight and obesity following migration to the UK, with consequences for biomarkers of inflammation. This might then override any potential protective effect of early life education of the immune system. In addition to central adiposity, generalized overweight and obesity are accompanied by chronic inflammation reflected in higher CRP levels [67–69]. The mechanisms whereby obesity leads to increased CRP levels are not clearly understood, but inflammatory cytokines such as IL-6 are produced by adipose tissue [68], and this process is exacerbated in conditions of increased central adiposity.

The relationship between gender and CRP levels is controversial across epidemiological studies. Data from European populations have shown no significant differences between females and males [70-72], but two large multi-ethnic studies reported that women had higher CRP levels irrespective of ethnic background [73, 74]. Studies of south-east Asian groups reported that Korean [75] and Japanese [74] men had higher CRP levels compared to women, while no gender differences were reported by Ye et al. for Chinese subjects [76]. Although the reasons for inconsistencies in gender difference in CRP concentrations are not clearly understood, differences in lifestyles and metabolic risk factors between men and women may be responsible. A meta-analysis of CRP demonstrated interactions between obesity and elevated CRP levels that were more marked among women than men, and also among populations from Europe and the United States [69].

In the study here, Gender was consistently a significant predictor of logCRP. When Models 1 and 2 were run separately by Gender, Generation was no longer a significant predictor for women but this was not the case for men, while WC remained significant in models for both. Klein and Flanagan's [64] observations, discussed above, of a dearth of reporting results by sex in the immunological literature is, in fact, mirrored in the relevant evolutionary literature. For example, the variable 'gender' was a significant predictor of CRP levels in the study of Ecuadorian migrants by Shattuck-Heidorn et al. [32], but McDade et al. [1] did not include this variable as a predictor in their model for the Cebu population in the Philippines. Our findings highlight the need to shed more light on sex differences in susceptibilities to inflammatory disorders among future such studies [64].

Alternatively, or in addition, there may be gendered differences in the ways in which middle class, Bangladeshi girls and boys are exposed to STH in early life, with potential for boys to have greater exposure if they are allowed more freedom to play outside or get dirty [64]. Existing data for infection with STH among Bangladeshi children in their home country are almost wholly confined to the poor whose lives are different from the middle classes. In just one paper, however, where children aged 0-15 years from different socioeconomic backgrounds and living conditions were compared [77], there were significant differences in prevalence of STH infection among the different groups ranging from 81% in children who lived in the worst housing to only 18% among those who lived in the highest quality housing. These differences were mirrored when children were grouped by family income. Hug and Shaikh [77] also noted that infections were higher among boys than girls. These data, and others for Bangladesh [78, 79], suggest that infections among children up to about 4-5 years old are higher for boys with the situation reversing between genders at about adolescence. These findings might imply that the immune system of boys is either more exposed to STH during early childhood, or reflects gender differences in the response to such exposures. Boys are also known to have much higher rates of respiratory infections than girls during childhood and these might also influence the ways in which the immunological system develops during early life [80].

In relation to how life course exposures might modify the potential educational effect on the immune system, Shattuck-Heidorn et al. [32] proposed from their findings of Ecuadorian migrants in the USA that the generally low levels of CRP in this population point to a life-long benefit of developmental regulation of immune function, but their sample was extremely small and included only one migrant group. In an earlier paper, McDade et al. [81] had compared their results for high-sensitivity CRP levels from the Cebu Longitudinal Study (when participants were aged 20-22) with data from 616 young adults aged 19-24 from the US National Health and Nutrition Examination Survey. The goal was to assess the impact of markers of adiposity, including WC, on levels of CRP in these populations. Overall, the Cebu sample had low plasma levels of baseline CRP, as described earlier in their 2010 study [1] and lower anthropometric indices compared to the US sample. In both the Cebu and US groups, both gender and WC predicted CRP, similar to findings in our study here. However, the Cebu sample had significantly lower plasma CRP levels for matched WC measurements compared to the US sample, even where these were comparatively high (>70 cm). In the Bangladeshi groups presented here, CRP levels were similar across the generations among individuals above the cut-off for central obesity, and it is only those individuals with a normal WC where significant differences emerged by developmental environment (Fig. 2).

One potential issue with McDade et al.'s [81] results was that the US samples were assayed in a different laboratory possibly introducing inter-laboratory error. The participants were also sufficiently young that it would be difficult to predict longerterm effects of obesity on CRP levels. Our study here, however, compared older individuals aged 25–40 from the same source population in Sylhet, northeast Bangladesh, some of whom had changed environments and had spent variable amounts of time in the UK. This age range is still young enough that increases in CRP would likely reflect effects of inflammatory conditions such as overweight/obesity as opposed to independent age effects [43, 44]. Indeed, the only group to show a positive correlation between age and CRP was the second-generation while, conversely, their BMI and WC were high but not correlated with age.

All samples were collected using the same protocols and were assayed in the same laboratory. While first-generation migrants had lower levels of salivary CRP compared to the second-generation overall, the data showed that overweight, and particularly high central obesity, appeared to mitigate the beneficial impact of the early life environment on CRP levels. Despite the effects of rising indicators of obesity, levels of CRP among Adult Migrants and Child Migrants >8, on average, were never as high as those among the second-generation British-Bangladeshis and Child Migrants  $\leq$ 8. These differences across the generations could provide valuable information to health practitioners when assessing health risks among Bangladeshis and South Asian migrants considered at greater risk for inflammatory disorders compared to some other ethnic groups.

It remains to be seen how rising levels of overweight and obesity within LMICs like Bangladesh will also affect levels of CRP across the life course, particularly among those who continue living in areas with high pathogen exposures. At present, Bangladesh like the Philippines and many other countries, suffers from the dual burden of morbidity with a variety of chronic diseases resulting from rapid development [82–84]. This development has been accompanied by lifestyle changes promoting obesogenic environments for which existing health systems are ill-prepared [85], and which can exist alongside persistent poverty [86–90].

Chronically elevated levels of CRP are an independent risk factor for several morbidities including CVDs [91, 92], T2DM [8], the metabolic syndrome [93], stress [94] overall frailty [95], and mortality among the elderly [4, 56], but are also symptomatic of several of these conditions. Relative to other South Asian countries like India and Pakistan, rates for T2DM in Bangladesh are higher; approximately 8% of the urban population and 2-3% of the rural population in Bangladesh were estimated to suffer from T2DM in 2006 [96], while absolute numbers of people with T2DM are expected to more than triple in Bangladesh by 2030 [97]. These morbidities are expected to impact significantly on health budgets [52]. Understanding the role of the childhood environment in influencing later life health, particularly in specific, early parts of childhood, alongside the role of gender in influencing environmental exposures, and a greater appreciation of evolutionary influences on our phenotypes, may therefore help in the development of strategies to mitigate these increasing health problems.

#### Study strengths and limitations

The study is cross-sectional and compared salivary CRP levels across Bangladeshi groups in north and south UK. All of the migrant population originated from Sylhet region in northeast Bangladesh, came from the same socioeconomic stratum and were uniformly Muslim, allowing for control of various cultural factors that might otherwise affect CRP levels. Participants were also drawn from limited geographic areas in London and northeast England thereby controlling for place of residence in the UK. Saliva samples were taken using the same sampling framework for all individuals and assayed in the same laboratory using standardized techniques. Collection of a large amount of qualitative and quantitative data allowed us to control for potential covariates while exploring potential explanatory variables.

The study relied on one sample for measurements of CRP rather than repeated samples which would have permitted intraindividual and more long-term comparisons, as advocated by McDade and others [98, 99]. However, at least in environments with low exposure to infectious diseases, levels of CRP are relatively stable within individuals [100]. Less is known about the validity of high specificity CRP levels in saliva compared to plasma, although such measures have been increasingly used in recent years as well as representing a less invasive method of data collection [101]. A majority of studies show good correlation (r=0.72) with high specificity plasma measures of CRP [15].

Given the labour-intensive methods of data collection for this study, participants were asked to mail their samples back to the laboratory. Some samples may therefore have been in transit slightly longer than others which may have affected sample quality. We attempted to control for this by only assaying the evening sample which would have been in transit for the least amount of time. Ouellet-Morin et al. [15] found a 12–14% decline in salivary CRP values across a 24– 96 h delay in freezing when samples were kept at room temperature, but potential waiting and transit times affected all our samples equally and there is unlikely to be any form of systematic bias between groups in this respect. It does mean, however, that levels of CRP reported here are likely to be lower overall than if we had been able to freeze samples immediately after collection.

#### SUPPLEMENTARY DATA

Supplementary data is available at EMPH online.

#### DATA ACCESSIBILITY

The data are available at the Durham Research Online Data Repository: doi:10.15128/r1gx41mh89z.

#### ACKNOWLEDGEMENTS

We are grateful to all participants and local contacts for their willingness to be part of the study and for sharing their valuable time. We also thank Mehdi Hasan and Raihana Ferdaus for assisting with data collection in the northeast, and Cllr Ayesha Chowdhury, Cllr Rohima Rahman, Lucky Miah, Razekul Islam, Rowshanara Chowdhury, Sheuli Khatun, Abdul Rahman and Prianta Alam for outreach to local communities in London.

#### FUNDING

Supported by the Economic and Social Research Council, UK (ES/J008478/1).

#### AUTHORS' CONTRIBUTIONS

GB, GC, AK and PN conceptualized the project; NA, KB, and PN collected the data; KB, GB, NA and AK performed the statistical analyses; GC analyzed the hormonal data; KB and GB wrote the initial drafts of the paper, and all authors contributed to subsequent drafts.

Conflict of interest: None declared.

#### REFERENCES

- McDade TW, Rutherford J, Adair L *et al.* Early origins of inflammation: microbial exposures in infancy predict lower levels of C-reactive protein in adulthood. *Proc R Soc B Biol Sci* 2010;277:1129–37.
- McDade TW, Tallman PS, Madimenos FC *et al.* Analysis of variability of high sensitivity C-reactive protein in lowland Ecuador reveals no evidence of chronic low-grade inflammation. *Am J Hum Biol* 2012;24: 675–81.
- Markanday A. Acute phase reactants in infections: evidence-based review and a guide for clinicians. Open Forum Infect Dis 2015;2:ofv098.
- Chambers JC, Eda S, Bassett P *et al.* C-reactive protein, insulin resistance, central obesity, and coronary heart disease risk in Indian Asians from the United Kingdom compared with European whites. *Circulation* 2001;**104**:145–50.
- Rexrode KM, Pradhan A, Manson JE *et al.* Relationship of total and abdominal adiposity with CRP and IL-6 in women. *Ann Epidemiol* 2003; 13:674–82.
- McDade TW, Rutherford JN, Adair L *et al.* Adiposity and pathogen exposure predict C-reactive protein in Filipino women. *J Nutr* 2008;138: 2442–7.
- Danesh J. Low grade inflammation and coronary heart disease: prospective study and updated meta-analyses. Br Med J 2000;321: 199–204.
- Pradhan AD, Manson JE, Rifai N *et al.* C-reactive protein, interleukin 6, and risk of developing type 2 diabetes mellitus. *JAMA* 2001;286:327.
- 9. Wang X, Bao W, Liu J *et al.* Inflammatory markers and risk of type 2 diabetes: a systematic review and meta-analysis. *Diabetes Care* 2013;**36**:166–75.
- Graham C, Chooniedass R, Stefura WP et al. Stability of pro- and antiinflammatory immune biomarkers for human cohort studies. J Transl Med 2017;15:53.

- Pearson TA, Mensah GA, Alexander RW *et al.*; American Heart Association. Markers of inflammation and cardiovascular disease. *Circulation* 2003;**107**:499–511.
- 12. Yeh ETH, Willerson JT. Coming of age of C-reactive protein. *Circulation* 2003;**107**:370–1.
- Roberts W, Moulton L, Law TC *et al.* Evaluation of nine automated high-sensitivity C-reactive protein methods: implications for clinical and epidemiological applications. Part 2. *Clin Chem* 2001;47:418–25.
- 14. Dizgah IM. Serum and saliva levels of high-sensitivity C-reactive protein in acute myocardial infarction. J Mol Biomark Diag 2012;03:3.
- 15. Ouellet-Morin I, Danese A, Williams B *et al.* Validation of a highsensitivity assay for C-reactive protein in human saliva. *Brain Behav Immun* 2011;**25**:640–6.
- Kramer A, Bekeschus S, Bröker BM *et al.* Maintaining health by balancing microbial exposure and prevention of infection: the hygiene hypothesis versus the hypothesis of early immune challenge. *J Hospital Infect* 2013;83:S29–34.
- Lisciandro JG, van den Biggelaar AHJ. Neonatal immune function and inflammatory illnesses in later life: lessons to be learnt from the developing world? *Clin Exp Allergy* 2010;40:1719–31.
- Rook GAW. 99th Dahlem Conference on Infection, Inflammation and Chronic Inflammatory Disorders: Darwinian medicine and the 'hygiene' or 'old friends' hypothesis. *Clin Exp Immunol* 2010;**160**:70–9.
- 19. Yazdanbakhsh M, Kremsner PG, van Ree R. Allergy, parasites, and the hygiene hypothesis. *Science* 2002;**296**:490–4.
- Furman D, Campisi J, Verdin E et al. Chronic inflammation in the etiology of disease across the life span. Nat Med 2019;25:1822-32.
- McDade TW. Life history theory and the immune system: steps toward a human ecological immunology. Am J Phys Anthropol 2003;122:100-25.
- 22. The ecologies of human immune function. *Annu Rev Anthropol* 2005;**34**:495–521.
- 23. Renz H, Adkins BD, Bartfeld S *et al*. The neonatal window of opportunity—early priming for life. J Allergy Clin Immunol 2018;**141**:1212–4.
- Ligthart S, Marzi C, Aslibekyan S *et al.*; CHARGE epigenetics of Coronary Heart Disease. DNA methylation signatures of chronic lowgrade inflammation are associated with complex diseases. *Genome Biol* 2016;**17**:255–70.
- 25. McDade TW, Ryan C, Jones MJ *et al.* Social and physical environments early in development predict DNA methylation of inflammatory genes in young adulthood. *Proc Natl Acad Sci USA* 2017;**114**:7611–6.
- 26. CIA. The World Factbook (Bangladesh). Washington, DC: Central Intelligence Agency, 2020.
- 27. Ahmed SM, Alam BB, Anwar I *et al.* Bangladeshi health system review. *Health Syst Transit* 2015;**5**:34.
- National Institute of Population Research and Training (NIPORT).
  2016. Bangladesh Demographic and Health Survey 2014. Dhaka, Bangladesh: Mitra and Associates and ICF International.
- Benjamin-Chung J, Nazneen A, Halder AK *et al.* The interaction of deworming, improved sanitation, and household flooring with soiltransmitted helminth infection in rural Bangladesh. *PLoS Neglected Trop Dis* 2015;**9**:e0004256.
- Dhakal S, Karim MJ, Kawsar A *et al.* Post-intervention epidemiology of STH in Bangladesh: data to sustain the gains. *PLoS Neglected Trop Dis* 2020;14:e0008597.
- Ministry of Health and Family Welfare G of B. Neglected Tropical Diseases in Bangadesh: Situation Analysis and Gap Analysis, 2011.

https://pdf.usaid.gov/pdf\_docs/pnady849.pdf (November 2021, date last accessed).

- Shattuck-Heidorn H, Eick GN, Kramer KL *et al.* Variability of C-reactive protein in first-generation Ecuadorian immigrants living in the United States. *Am J Hum Biol* 2021;**33**:e23547.
- Urlacher SS, Ellison PT, Sugiyama LS *et al.* Tradeoffs between immune function and childhood growth among Amazonian forager-horticulturalists. *Proc Natl Acad Sci USA* 2018;**115**:E3914–21.
- Bhopal RS. A four-stage model explaining the higher risk of Type 2 diabetes mellitus in South Asians compared with European populations. *Diabetic Med* 2013;30:35–42.
- 35. Khan MN, Rahman MM, Shariff AA *et al.* Maternal undernutrition and excessive body weight and risk of birth and health outcomes. *Arch Public Health* 2017;**75**:12.
- McKeigue PM, Shah B, Marmot MG. Relation of central obesity and insulin resistance with high diabetes prevalence and cardiovascular risk in South Asians. *Lancet* 1991;337:382–6.
- 37. Misra A, Chowbey P, Makkar BM *et al.*; Concensus Group. Consensus statement for diagnosis of obesity, abdominal obesity and the metabolic syndrome for Asian Indians and recommendations for physical activity, medical and surgical management. *J Assoc Physicians India* 2009;**57**: 163–70.
- 38. Akhter N, Begum K, Nahar P et al. Risk factors for non-communicable diseases related to obesity among first- and second-generation Bangladeshi migrants living in north-east or south-east England. Int J Obesity 2021;45:1588–98.
- 39. Núñez-De La Mora A, Chatterton RT, Choudhury OA *et al.* Childhood conditions influence adult progesterone levels. *PLoS Med* 2007;4:e167.
- Begum K, Muttukrishna S, Sievert LL *et al.* Ethnicity or environment: effects of migration on ovarian reserve among Bangladeshi women in the United Kingdom. *Fertil Steril* 2016;**105**:744–54.e1.
- Magid K, Chatterton RT, FU A *et al.* Childhood ecology influences salivary testosterone, pubertal age and stature of Bangladeshi UK migrant men. *Nat Ecol Evol* 2018;2:1146–57.
- Murphy L, Sievert L, Begum K *et al.* Life course effects on age at menopause among Bangladeshi sedentees and migrants to the UK. *Am J Hum Biol* 2013;25:83–93.
- Puzianowska-Kuźnicka M, Owczarz M, Wieczorowska-Tobis K et al. Interleukin-6 and C-reactive protein, successful aging, and mortality: the PolSenior study. *Immun Ageing* 2016;13:21. 10.1186/s12979-016-0076-x.
- Tang Y, Fung E, Xu A, Lan H-Y. C-reactive protein and ageing. *Clin Exp* Pharmacol Physiol 2017;44:9–14. 10.1111/1440-1681.12758.
- 45. Lohmann RG, Roche AF, Martorell R. 1988. Anthropometric Standardization Reference Manual. Champaign, IL: Human Kinetics Books.
- WHO. Body Mass Index BMI. 2020. http://www.euro.who.int/en/ health-topics/disease-prevention/nutrition/a-healthy-lifestyle/bodymass-index-bmi. (November 2021, date last accessed).
- NICE. Obesity: identification, assessment and management Clinical guideline [CG189]. 2014. https://www.nice.org.uk/guidance/cg189. (November 2021, date last accessed).
- Núñez-de la Mora A, Chatterton RT, Mateo ET *et al.* Effect of chewing betel nut on measurements of salivary progesterone and estradiol. *Am* J Phys Anthropol 2007;132:311–5.
- Goldberg, DP, Williams, P. A User's Guide to the General Health Questionnaire. NFER-Nelson, 1988.

- Suinn RM, Ahuna C, Khoo G. The Suinn-Lew Asian Self-Identity Acculturation Scale: concurrent and factorial validation. *Educ Psychol Meas* 1992;52:1041–6.
- Bloomfield SF, Stanwell-Smith R, Crevel RWR *et al.* Too clean, or not too clean: the Hygiene Hypothesis and home hygiene. *Clini Exp Allergy* 2006;**36**:402–25.
- 52. McDade TW. Early environments and the ecology of inflammation. *Proc Natl Acad Sci USA* 2012;**109**:17281–8.
- Blackwell AD, Snodgrass JJ, Madimenos FC *et al.* Life history, immune function, and intestinal helminths: trade-offs among immunoglobulin E, C-reactive protein, and growth in an Amazonian population. *Am J Hum Biol* 2010;**22**:836–48.
- Tanchoco CC, Cruz AJ, Duante CA *et al.* Prevalence of metabolic syndrome among Filipino adults aged 20 years and over. *Asia Pacific J Clin Nutr* 2003;12:271–6.
- 55. Bellary S, Paul O'Hare J, Raymond NT *et al.* Premature cardiovascular events and mortality in south Asians with type 2 diabetes in the United Kingdom Asian Diabetes Study—effect of ethnicity on risk. *Curr Med Res Opin* 2010;**26**:1873–9.
- Bhopal RS, Rafnsson SB, Agyemang C et al. Mortality from circulatory diseases by specific country of birth across six European countries: test of concept. Eur J Public Health 2012;22:353–9.
- Schwartz BS, Harris JB, Khan AI *et al.* Diarrheal epidemics in Dhaka, Bangladesh during three consecutive floods. *Am J Trop Med Hyg* 2006; 74:1067–73.
- 58. UNICEF. Towards Better Health: The Rural Water Supply and Sanitation Programme in Bangladesh. 1992. https://fr.ircwash.org/ sites/default/files/822-BD95-13659.pdf. (November 2021, date last accessed).
- Bhuiyan MAI, Jhinu ZN, Jaliluzzaman *et al.* Personal hygiene practices and socio-economic conditions as influential factors for intestinal parasitic infection in Dhaka city dwellers. *Bangladesh J Zool* 2019;47: 129–36.
- WHO. Factsheet 2020 Bangladesh. Expanded Programme on Immunization (EPI). 2020. https://apps.who.int/iris/handle/10665/ 336754. Last accessed November 2021.
- Sarkar P, Sarkar N, Doulah S *et al.* Expanded programme on immunization in Bangladesh: a success story. *Bangladesh J Child Health* 2017;**39**: 93–8.
- 62. Wardrop NA, Hill AG, Dzodzomenyo M *et al.* Livestock ownership and microbial contamination of drinking-water: evidence from nationally representative household surveys in Ghana, Nepal and Bangladesh. *Int* J Hyg Environ Health 2018;**221**:33–40.
- 63. Klein SL, Flanagan KL. Sex differences in immune responses. *Nat Rev Immunol* 2016;**16**:626–38.
- 64. Clough S. Gender and the hygiene hypothesis. *Soc Sci Med* . 2011;**72**: 486–493.
- 65. Brooks GC, Blaha MJ, Blumenthal RS. Relation of C-reactive protein to abdominal adiposity. *Am J Cardiol* 2010;**106**:56–61.
- 66. Shemesh T, Rowley KG, Jenkins A *et al.* Differential association of C-reactive protein with adiposity in men and women in an Aboriginal community in northeast Arnhem Land of Australia. *Int J Obes* 2007;**31**: 103–8.
- 67. Alley DE, Seeman TE, Ki Kim J *et al.* Socioeconomic status and C-reactive protein levels in the US population: NHANES IV. *Brain Behav Immun* 2006;**20**:498–504.

- 68. Cancello R, Clément K. Review article: is obesity an inflammatory illness? Role of low-grade inflammation and macrophage infiltration in human white adipose tissue. BJOG Int J Obstetr Gynaecol 2006;113:1141–7.
- Choi J, Joseph L, Pilote L. Obesity and C-reactive protein in various populations: a systematic review and meta-analysis. *Obes Rev* 2013;14: 232–44.
- 70. Araújo F, Pereira AC, Latorre M. d R *et al.* High-sensitivity C-reactive protein concentration in a healthy Brazilian population. *Int J Cardiol* 2004;**97**:433–8.
- García-Lorda P, Bulló M, Balanzà R *et al.* C-reactive protein, adiposity and cardiovascular risk factors in a Mediterranean population. *Int J Obes* 2006;**30**:468–74.
- Imhof A, Fröhlich M, Loewel H *et al.* Distributions of C-reactive protein measured by high-sensitivity assays in apparently healthy men and women from different populations in Europe. *Clin Chem* 2003;49: 669–72.
- 73. Lakoski SG, Cushman M, Criqui M *et al*. Gender and C-reactive protein: data from the Multiethnic Study of Atherosclerosis (MESA) cohort. *Am Heart J* 2006;**152**:593–8.
- 74. Yamada S, Gotoh T, Nakashima Y *et al.* Distribution of serum C-reactive protein and its association with atherosclerotic risk factors in a Japanese population: Jichi Medical School Cohort Study. *Am J Epidemiol* 2001;**153**:1183–90.
- Lee Y-J, Lee J-H, Shin Y-H *et al.* Gender difference and determinants of C-reactive protein level in Korean adults. *Clin Chem Lab Med* 2009;47: 863–9.
- 76. Ye X, Yu Z, Li H *et al.* Distributions of C-reactive protein and its association with metabolic syndrome in middle-aged and older Chinese people. J Am Coll Cardiol 2007;49:1798–805.
- Huq M, Shaikh A. Incidence of intestinal parasite in children of different socio-economic population of Dhaka City, Bangladesh. *Bangladesh Med Res Council Bull* 1976;11:20–6.
- Khanum H, Nahar A, Karim MT *et al.* Infection of protozoan and helminth parasites among the out-patients of Dhaka Medical College Hospital. *Bangladesh J Zool* 2016;44:89–97.
- Khanum H, Rahman MM, Uddin MH *et al.* Intestinal parasitic infestation among the outdoor patients of Dhaka University Medical Centre, Bangladesh. Univ J Zool Rajshahi University 1970;28:45–9.
- Nair H, Simõ EAF, Rudan I *et al.* Articles Global and regional burden of hospital admissions for severe acute lower respiratory infections in young children in 2010: a systematic analysis. *Lancet* 2013;381: 1380–90.
- McDade TW, Rutherford JN, Adair L *et al.* Population differences in associations between C-reactive protein concentration and adiposity: comparison of young adults in the Philippines and the United States. *Am J Clin Nutr* 2009;89:1237–45.
- 82. Alam DS, Robinson H, Kanungo A *et al.* Health system preparedness for responding to the growing burden of non-communicable disease a case study of Bangladesh. The Nossal Institute for Global Health. University of Melbourne, Working Paper Series, 2013, 25.
- Bleich SN, Koehlmoos TLP, Rashid M et al. Noncommunicable chronic disease in Bangladesh: overview of existing programs and priorities going forward. *Health Policy* 2011;**100**:282–9.
- Hoque ME, Long KZ, Niessen LW *et al.* Rapid shift toward overweight from double burden of underweight and overweight among Bangladeshi women: a systematic review and pooled analysis. *Nutr Rev* 2015;**73**:438–47.

- Islam A. Health system in Bangladesh: challenges and opportunities. *Am J Health Res* 2014;2:366.
- Biswas T, Islam MS, Linton N *et al.* Socio-economic inequality of chronic non-communicable diseases in Bangladesh. *PLoS One* 2016;11: e0167140.
- Rahim MA, Hussain A, Azad Khan AK et al. Rising prevalence of type 2 diabetes in rural Bangladesh: a population based study. *Diabetes Res Clin Pract* 2007;77:300–5.
- Rahman S, Islam MT, Alam DS. Obesity and overweight in Bangladeshi children and adolescents: a scoping review. *BMC Public Health* 2014; 14:70.
- 89. Saquib N, Saquib J, Ahmed T et al. Cardiovascular diseases and type 2 diabetes in Bangladesh: a systematic review and metaanalysis of studies between 1995 and 2010. BMC Public Health 2012;12:434.
- 90. Forouhi NG, Sattar N, Tillin T *et al.* Do known risk factors explain the higher coronary heart disease mortality in South Asian compared with European men? Prospective follow-up of the Southall and Brent studies, UK. *Diabetologia* 2006;**49**:2580–8.
- Albert CM, Ma J, Rifai N *et al.* Prospective study of C-reactive protein, homocysteine, and plasma lipid levels as predictors of sudden cardiac death. *Circulation* 2002;**105**:2595–9.
- Stoner L, Lucero AA, Palmer BR et al. Inflammatory biomarkers for predicting cardiovascular disease. Clin Biochem 2013;46:1353–71.

- Ridker PM, Buring JE, Cook NR *et al.* C-reactive protein, the metabolic syndrome, and risk of incident cardiovascular events. *Circulation* 2003; 107:391–7.
- 94. Steptoe A, Hamer M, Chida Y. The effects of acute psychological stress on circulating inflammatory factors in humans: a review and meta-analysis. *Brain Behav Immun* 2007;**21**:901–12.
- 95. Jenny NS, Yanez ND, Psaty BM *et al.* Inflammation biomarkers and near-term death in older men. *Am J Epidemiol* 2007;**165**:684–95.
- Shaw JE, Sicree RA, Zimmet PZ. Global estimates of the prevalence of diabetes for 2010 and 2030. *Diabetes Res Clin Pract* 2010;87:4–14.
- 97. Bishwajit G. Nutrition transition in South Asia: the emergence of noncommunicable chronic diseases. *F1000Res* 2015;**4**:8.
- Macy EM, Hayes TE, Tracy RP. Variability in the measurement of C-reactive protein in healthy subjects: implications for reference intervals and epidemiological applications. *Clin Chem* 1997;43:52–8.
- 99. Ockene IS, Matthews CE, Rifai N *et al.* Variability and classification accuracy of serial high-sensitivity C-reactive protein measurements in healthy adults. *Clin Chem* 2001;**47**:444–50.
- 100. Slavish DC, Graham-Engeland JE, Smyth JM *et al.* Salivary markers of inflammation in response to acute stress. *Brain Behav Immun* 2015; 44:253–69.
- 101. Megson E, Fitzsimmons T, Dharmapatni K et al. C-reactive protein in gingival crevicular fluid may be indicative of systemic inflammation. J Clin Periodontol 2010;37:797–804.